

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Wübker, Ansgar

Working Paper Explaining Variations in Breast Cancer Screening Across European Countries

Ruhr Economic Papers, No. 370

Provided in Cooperation with: RWI – Leibniz-Institut für Wirtschaftsforschung, Essen

Suggested Citation: Wübker, Ansgar (2012) : Explaining Variations in Breast Cancer Screening Across European Countries, Ruhr Economic Papers, No. 370, ISBN 978-3-86788-425-9, Rheinisch-Westfälisches Institut für Wirtschaftsforschung (RWI), Essen, https://doi.org/10.4419/86788425

This Version is available at: https://hdl.handle.net/10419/67125

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



RUHR ECONOMIC PAPERS

Ansgar Wübker



Imprint

Ruhr Economic Papers

Published by

Ruhr-Universität Bochum (RUB), Department of Economics Universitätsstr. 150, 44801 Bochum, Germany

Technische Universität Dortmund, Department of Economic and Social Sciences Vogelpothsweg 87, 44227 Dortmund, Germany

Universität Duisburg-Essen, Department of Economics Universitätsstr. 12, 45117 Essen, Germany

Rheinisch-Westfälisches Institut für Wirtschaftsforschung (RWI) Hohenzollernstr. 1-3, 45128 Essen, Germany

Editors

Prof. Dr. Thomas K. Bauer
RUB, Department of Economics, Empirical Economics
Phone: +49 (0) 234/3 22 83 41, e-mail: thomas.bauer@rub.de
Prof. Dr. Wolfgang Leininger
Technische Universität Dortmund, Department of Economic and Social Sciences
Economics - Microeconomics
Phone: +49 (0) 231/7 55-3297, email: W.Leininger@wiso.uni-dortmund.de
Prof. Dr. Volker Clausen
University of Duisburg-Essen, Department of Economics
International Economics
Phone: +49 (0) 201/1 83-3655, e-mail: vclausen@vwl.uni-due.de
Prof. Dr. Christoph M. Schmidt
RWI, Phone: +49 (0) 201/81 49-227, e-mail: christoph.schmidt@rwi-essen.de

Editorial Office

Joachim Schmidt RWI, Phone: +49 (0) 201/81 49-292, e-mail: joachim.schmidt@rwi-essen.de

Ruhr Economic Papers #370

Responsible Editor: Thomas K. Bauer

All rights reserved. Bochum, Dortmund, Duisburg, Essen, Germany, 2012

ISSN 1864-4872 (online) - ISBN 978-3-86788-425-9

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editors.

Ruhr Economic Papers #370

Ansgar Wübker

Explaining Variations in Breast Cancer Screening Across European Countries



Bibliografische Informationen der Deutschen Nationalbibliothek

Die Deutsche Bibliothek verzeichnet diese Publikation in der deutschen Nationalbibliografie; detaillierte bibliografische Daten sind im Internet über: http://dnb.d-nb.de abrufbar.

http://dx.doi.org/10.4419/86788425 ISSN 1864-4872 (online) ISBN 978-3-86788-425-9 Ansgar Wübker¹

Explaining Variations in Breast Cancer Screening Across European Countries

Abstract

This paper explores variations in the uptake of breast cancer screening and associated factors influencing utilisation of mammography screening among women aged 50 to 69 years in 13 European countries. We focus on the relative importance of individual (e.g. age, education, etc.) and institutional (e.g. public screening program) factors in explaining cross-country variation in the utilisation of mammograms. We take advantage of (a) newly available individual level data from the SHARE as well as (b) regional and country level data on institutional factors. We find that observed individual factors like age, education, health status, etc. are associated with screening uptake within countries but cannot statistically explain cross-country differences. In contrast, observed institutional factors like the availability of an organized screening program can statistically explain about 40 per cent of the between country differences in screening rates.

JEL Classification: C01, I11, I18

Keywords: Health economics; prevention; multilevel models; SHARE; cross country differences

September 2012

¹ Ruhr-Universität Bochum and Department of Economics, Witten/Herdecke University. - The author thanks Marc Dusheiko, Jens Harbecke, Jan Kleibrink, Jürgen Maurer, Maren M. Michaelsen and all participants of the PhD seminar on health economics and policy in Grindelwald for valuable comments. All the remaining errors are mine. This paper uses data from SHARELIFE release 1, as of November 24th 2010 or SHARE release 2.3.1, as of July 29th 2010. The SHARE data collection has been primarily funded by the European Commission through the 5th framework programme (project QLK6-CT-2001- 00360 in the thematic programme Quality of Life), through the 6th framework programme (projects SHARE-13, RII-CT- 2006-062193, COMPARE, CIT5-CT-2005-028857, and SHARELIFE, CIT4-CT-2006-028812) and through the 7th framework programme (SHARE-PREP, 211909 and SHARE-LEAP, 227822). Additional funding from the U.S. National Institute on Aging (Uo1 AG09740-13S2, P01 AG005842, P01 AG08291, P30 AG12815, Y1-AG-4553-01 and OGHA 04-064, IAG BSR06-11, R21 AG025169) as well as from various national sources is gratefully acknowledged (see www.share-project.org/t3/share/index.php for a full list of funding institutions). - All correspondence to Ansgar Wübker, Ruhr-Universität Bochum, Chair of Competition Theory and Policy, Universitätsstr. 150, 44801 Bochum, Germany, E-mail: ansgar. wuebker@rub.de.

1 Introduction

Breast cancer is the most common cause of cancer death in the member states of the European Union (von Karsa et al., 2008). According to estimates of incidence and mortality by the International Agency for Research on Cancer (IARC), there were 331,000 new cases and 90,000 deaths due to breast cancer in the EU in 2006 (von Karsa et al., 2008). Breast cancer accounts for almost one out of three new cancer cases and one out of six (17 per cent) cancer deaths. One in nine women gets breast cancer at some point in her life and one in thirty dies from the disease (OECD, 2009). Due to demographic trends a significantly higher proportion of women will be confronted with breast cancer in the future (Ferlay et al., 2007). Moreover, breast cancer is associated with high costs for national health care. Overall spending for breast cancer typically amounts to about 0.5-0.6 per cent of the total health care expenditure of developed countries (OECD, 2009). Breast cancer typically takes years to develop. At the onset of the disease, most breast cancers cause no symptoms. As long as the cancer has not metastasized, i.e. has not moved to the lymph system or to other organs of the body, patients have a five-year survival rate of 96 per cent. If the cancer has spread to the nearby lymph nodes, the rate drops down to 81 per cent. Women whose breast cancer has metastasized to other organs of the body have a five-year survival rate of 26 per cent (Fang and Wang, 2010). Mammogram screening is the best tool available for detecting breast cancer in the early stage, i.e. before symptoms appear. Mammography can detect a breast lump before it can be palpated; it can save lives by detecting breast cancer in the earliest stage. On the basis of several randomized clinical trials, the World Health Organization concluded in 2002 that in areas with screening attendance of at least 70 per cent, a reduction in breast cancer mortality by about 25 per cent may be expected in women screened between ages of 50 and 69 years.² In light of this evidence, the International Agency for Research on Cancer expert working group (IARC, 2002) advises that mammography screening should be offered as a public health policy directed to women aged 50-69 every two years in order to reduce the risk of death from breast cancer and also EU guidelines and national guidelines promote regular screening (i.e. every two years) for women of these ages (see Wübker, 2011 for further details).

² Compare for an overview (IARC 2002).

Despite these public health efforts to promote screening uptake, the use of screening mammography is not without controversy. Critics point to methodological limitations in some of the randomized trials (Kalager *et al.*, 2010) which might lead to overestimation of the effectiveness of screening.³ The discussion amongst the critics and advocates of breast cancer screening titled the "Mammography Wars" by Quanstrum and Hayward (2010) might be one reason why screening rates in most European countries remain far from 100 per cent and differ much between European countries as shown by Figure 1.

Insert Figure 1 here

As argued by Skinner (2012) or Chandra, Cutler and Song (2012) the largest regional variations in treatment can be found in "supply-sensitive" care or "grey area medicine", where clinical value of treatment is debated and thus in which clinical judgment plays a key role.⁴

Following Bolin *et al.* (2008) and building on the "geographic variation literature" (Skinner (2012) or Chandra, Cutler and Song (2012)), differences in healthcare utilization might be explained by (i) differences in demographic and socioeconomic composition of the population (individual factors), (ii) differences in health system regulation, financing and provision (explicit institutional factors) and (iii) differences in culture, tradition and norms (implicit institutional factors).

In this paper we focus on individual and explicit institutional factors and do so by laying out some theoretical and empirical arguments for the association between these factors and mammogram uptake. There exists a considerable amount of empirical and theoretical research in health economics on the predictors of screening and preventive behaviour. Theoretical economic models include those of Grossman (1972), Cropper (1977), Giuffrida and Gravelle (1998), Byrne and Thompson (2001), Howard (2005) or Fang and Wang (2010). Jepson et al. (2000) and Schueler *et al.* (2008) provide good reviews of the empirical literature on determinants of mammography screening uptake and recommendations for increasing uptake. While there is considerable evidence on factors associated with mammography screening, the reasons for country differences in screening remain unclear. This

³ Especially the omission of harms of screening due to false positive diagnosis and overdiagnoses is criticized by different authors (e.g. Raftery and Chorozoglu (2011)).

⁴ For a definition of "gray area medicine" compare Chandra, Cutler and Song (2012)

paper advances the literature on the determinants of screening uptake by providing first time empirical evidence on the factors statistically explaining country differences. Awareness of the statistical causes behind these country differences might provide useful information for the design of appropriate public health policies for influencing mammography uptake within countries.

To analyse the individual and institutional causes of screening differences across 13 European countries, we take advantage of (a) newly available internationally comparable individual level data which we combine with (b) regional level data on institutional factors. Individual level data are taken from the first three waves of the Survey of Health, Ageing and Retirement in Europe (SHARE). SHARE is a representative micro data set for the Europeans aged 50+ years including comprehensive information on health status, health risk, socioeconomic factors and mammography screening behaviour. We merge these data with country and regional level information on different institutional factors (e.g. relative number of mammography units or radiologists, existence of an organized screening program, age groups included in the screening program, intensity of screening program) which are taken from different sources (e.g. European Commission; von Karsa et al. 2008) as explained in the bottom of Table 2. Plausibly, the most important institutional factor influencing screening uptake is whether a region has introduced an organized population based screening program. Within an organized screening program eligible women (i.e. women aged 50-69 years) typically get regular (i.e. every two years) personal invitations to participate in free mammography screening at a location nearby imposing strong (economic) incentives fostering screening uptake. Incentives for screening uptake are even strengthened by the fact that within an organized screening program the screening process is accompanied by a comprehensive quality assurance. Moreover, women receive an information booklet which explains the pros and cons of mammography screening. In Europe we see enormous regional differences, across and within countries, in the availability of such screening programs and we exploit these differences to identify the impact of organized screening on screening uptake.

The analytical approach is based on multilevel statistical models. These models allow us to decompose the total variance in mammography screening rates into fractions that are due to differences in individual and regional factors within countries and due to differences in individual, regional and country factors between countries. These models enable us to statistically explain the variation (inequality) in screening uptake within a country as well as the differences in screening rates across countries.

We find the following main results: First, observed individual factors like age, income, education, health status, etc. can statistically explain screening uptake within countries. However, despite of considerable differences in the means for almost each individual factor across the countries, they cannot explain differences in uptake rates across countries. Second, in contrast, observed institutional factors like the availability of an organized screening program can statistically explain about 40 per cent of the between country differences in screening rates.

The remainder of the paper is organized as follows: Section 2 illustrates the theoretical and empirical screening determinants. Section 3 presents some information on the data sets. Section 4 explains the empirical strategy and section 5 presents the results. Section 6 discusses the results and adds some concluding remarks.

2 Mammography screening and its determinants

Individual factors

Differences in mammography screening within and across countries can be explained both by differences in individual and institutional factors, which affect demand and supply of screening. Michael Grossman's model of the demand for health care provides a useful framework for understanding the demand side, i.e. an individual's demand of preventive health care services. In this model, the decision to undergo mammography screening is an investment decision. Such an investment is worthwhile if the expected present value of the reduction in disease and in the probability of death is larger than the opportunity costs of the intervention (comp. Grossman, 1972, Cropper, 1977, Dardanoni and Wagstaff, 1990 or Chang, 1996 for a formalization of these notions). Existing empirical studies – motivated by economic demand theory – found that individual factors like

better education, better cognitive abilities, higher income, having a partner, younger age, a healthy lifestyle, and better health are associated with higher rates of receipt of screening (e.g. Lairson *et al.*, 2005, Wübker, 2011).

Alternatively, supply side factors have an important impact screening behaviour. Agency theory (e.g. McGuire, 2000) suggests that physician behaviour (i.e. whether a physician recommends screening) can be expected to influence the decision for screening, since asymmetric information is particularly widespread in health care markets often forcing expert physicians to act on behalf of their less informed patients. Moreover, individual perception of risks is often biased (e.g. Viscusi, 1990). Breast cancer is no exception in this regard and even women with a high risk of getting breast cancer may have false perceptions of the risks and the seriousness of breast cancer (Richards *et al.*, 2010). For this reason, physicians often need to act as agents for their less-informed patients, and they play an important role in determining mammography screening take-up.⁵ Empirical evidence clearly indicates that women tend to follow their physician's advice for mammography screening (e.g. May *et al.*, 1999, Meissner *et al.*, 2008). For example, May *et al.* (1999) find in an US-study that 66 per cent of women who received a recommendation adhered and of women receiving a documented recommendation, 75 per cent adhered. Alternatively Meissner *et al.* (2008) found for the US that 80 per cent of non-screeners who reported having access to healthcare did not receive a recommendation for a mammogram.

Institutional factors

The Grossman model follows an individual level approach wherein choices are made to maximise discounted lifetime utility subject to a number of monetary, time and other health systems constraints (Grossman, 1972). In this model, changes in the constraints have influence on individual behaviour. Indeed, empirical studies reveal that public health interventions within a health system which influence financing, delivery, etc. and therefore change the constraints of an individual have a strong impact on screening rates (for an overview compare Peek and Han, 2004). For example, a meta-

⁵ An additional theoretical supply-side approach to explaining geographic variations arises by allowing the production function to differ across regions or physicians. Compare Skinner (2012).

analysis (Legler *et al.*, 2002) shows that demand side interventions that enhance access (e.g. mobile vans, transportation services), reduce perceived costs (e.g. coverage of mammography by insurance) or influence perceived benefits of screening (e.g. community education, or mass media campaigns, introduction of guidelines) were all very successful in increasing mammography use within a country. Moreover researchers identified that supply side interventions like providing clear and non-conflicting guidelines for providers (e.g. Yasmeen *et al.*, 2012),⁶ bonus payments (i.e. Pay for Performance) for providing mammograms (e.g. Li *et al.*, 2011) or effort to change mammography prescribing behaviour of the physicians by reminder systems/office prompts (e.g. Mandelblatt and Yabroff, 1999) have shown to be effective and increased significantly mammography use.⁷

An organized population based screening program is a comprehensive public health intervention that combines many of the interventions just mentioned. Within an organized screening program eligible women (i.e. women aged 50-69 years) get regular (i.e. every two years) personal invitations to participate in free mammography screening.⁸ Moreover the screening is offered at a location nearby, women receive an information booklet which explains the pros and cons of mammography screening and screening is accompanied by comprehensive quality assurance for equipment, radiographers (technicians) as well as radiologists.

From an economic point of view and in accordance with the Grossman model and agency theory (e.g. McGuire, 2002) an organized screening program should increase mammography uptake since it changes the constraints for the woman concerned and physician involved. It provides incentives in terms of reduced access, information, transaction as well as time costs and plausibly increase perceived benefits of mammography screening. The introduction of an organized screening program

⁶ The authors show for the US that physicians consistently recommended mammography to women aged 50-69 years (the age group where leading professional societies consistently recommend screening) whereas widely differ with recommendation for younger women (e.g. aged between 40 and 49 years) and older women (over 75 years) where leading professional societies in the US issue conflicting recommendations.

⁷ Additionally Hamblin (1991) found that physician's recommendation of mammography was much lower when the mammography unit was far away.

⁸ In the EU-member states the characteristics of organized screening programs are widely generalizable, since the Commission of the European Communities succeeds to ensure a strict evidence base for implementing screening programs and the recognition of EU guidelines on best practice in the EU-member states (Com 2008). The IARC (2005) has defined the elements of an organized screening program. These include: 1) an explicit policy with specified age categories, method and interval for screening; 2) a defined target population; 3) a management team responsible for implementation; 4) a healthcare team for decisions and care; 5) a quality assurance structure; and 6) a method for identifying cancer occurrence in the target population.

plausibly "sends a message to providers" regarding the benefits of screening which might change physicians priorization, targets and incentives making them more prone to recommend screening. Compared to the alternative of opportunistic screening,⁹ organized screening puts a much greater emphasis on the quality of the screening process as measured by factors such as, tumour characteristics, cancer detection rates and false-positive biopsy rates (Miles *et al.*, 2004). Thus, it provides better protection against the harms of screening—including overscreening, poor quality, and complications of screening—and poor follow-up of those who test positively (Levin *et al.*, 2012, Miles *et al.*, 2004.) Accordingly, Strumpf, Chai and Kadiyala (2010) found that the implementation of an organized mammography screening program in Quebec strongly increased adherence to cancer screening guidelines and screening uptake.¹⁰

3 Data

We use data from the first (2004), second (2006) and third (2009) wave of the Survey of Health, Ageing and Retirement in Europe (SHARE) to explore variations in the uptake of breast cancer screening and associated factors influencing utilisation of mammography screening among women aged 50 to 69 years in 13 European countries. SHARE is a large representative micro data set providing detailed information on health, healthcare use, as well as a variety of other socioeconomic characteristics of more than 30 000 individuals above the age of 50 years starting in 2004.¹¹ Using the third wave of the SHARE (SHARELIFE) we calculate our dependent variable. SHARELIFE focuses on people's life histories and contains detailed information on historical mammography

⁹ Opportunistic screening happens when someone asks their doctor or health professional for a mammogram.

¹⁰ The authors measured compliance as the change in age-specific screening rates at the guideline-recommended initiation age (i.e. women older than 50 years). The authors found that after adjusting for age trends and other covariates, being above age 50 in Quebec, which is the only province where an organized screening program existed, increased the probability of being screened by 19 percentage points, from an average screening rate of 24% among 40-49 year old women. None of the other regions exhibited a statistically significant change in screening rates at age 50.

¹¹ For comprehensive information on the sampling procedure, questionnaire contents, and fieldwork methodology see Börsch-Supan and Jürges (2005).

screening use. The dependent variable measures whether a woman had regularly a mammogram at least every two years before 2007.¹²

In this paper, we restrict the sample to women aged 50-69, since for this group mammography screening is officially recommended at both European level and the national level of the countries included. Therefore, the estimation sample consists of 11,409 observations in total. Table 1 provides descriptive statistics for the individual variables included in the empirical analysis for the sample as a whole as well as for each country separately. A more detailed description of the variables included and the exact specification can be found in Table A1 of the Appendix. Explanatory variables include individual factors like age (dummy variables for different age groups), health status (SAH and number of ADL), a history of breast cancer, family structure (having a partner), income, education (as measured by ISCED-groups), cognitive abilities (as measured by verbal fluency), working, health behaviour and risk factors (taking medications). The rationale for including them is described in more detail in Wübker (2011). As Table 1 indicates, there are substantial differences in the means for almost each individual factor across the countries, which make it important to control for individual factors, when analysing the impact of differences in institutions (see Bolin et al., 2009). For example, the average age varied from 61 years in Poland to 64.4 years in Austria.

Insert Table 1 here

Table 2 provides information for the institutional variables included in the empirical analysis. The data refer to the situation in the years 2004 as well as 2006 and are derived from different international sources (compare bottom of Table 2). Of special interest are the variables that refer to the availability and organization of screening programs in the different European countries. The most important variable includes whether an organized screening program exists, since it imposes strong (economic) incentives fostering screening uptake (compare section 2). In contrast to the other institutional variables, which are only available on country level, we can also take advantage of regional variation

¹² For calculation of our dependent variable we use following questions: 1. Have you ever had mammograms regularly over the course of several years?; 2. In which year did you start having mammograms regularly?; 3. When you were having mammograms, how often was that on average? (Response options: 1. At least once a year; 2. Not every year, but at least every two years; 3. Less often); Since then, have you always had mammograms regularly?

of the availability of an organized screening program within countries (Nuts 2 code level). Some countries implemented an organized screening program in all regions (e.g. Sweden, Netherlands), whereas other countries offered it only in some regions (Italy, Denmark, Switzerland, Germany) and other countries (e.g. Poland, Greece) did not implement it at all.

Further explanatory variables regarding the organization of screening programs include, i) the share of women who are personally invited, ii) the screening interval in years, iii) the year in which the first screening program started, iv) the type (opportunistic versus organized) and geographical spreading (national versus regional) of the screening programs, v) whether the screening program is extended to age groups beyond women aged 50 to 69 years (e.g. 40 to 75 years), vi) the number of radiologists (i.e. the physicians who perform screening) per million women and vii) the number of mammography units per million women. Higher invitation rates, a shorter screening interval, a broader geographical spreading, an extended age range as well as higher numbers of radiologists and mammography units are c.p. all expected to lead to higher levels of utilisation, because they should lower information, transaction and access costs of screening. Moreover, Table 2 provides other regulatory (e.g. physician payment) and demand side (i.e. incidence of breast cancer) differences than the ones mentioned above which may affect mammography uptake. The data in Table 2 indicate that there are remarkable differences in the availability and organization of screening programs across the countries.

Insert Table 2 here

Finally, we use data on the reasons why women did not regularly take-up a mammography screening. SHARELIFE explores the causes why women had never had, or stopped having mammograms done regularly by focusing on perceived benefits, information issues, financial restrictions, time costs and availability of mammography services. Figure 2 reveals that there are enormous differences between countries in the self-stated reasons for not participating in screening. Regarding "Not considered to be necessary" 47 per cent of women in Denmark mention this factor as a cause for not getting a mammogram and only about 4 per cent of women in Sweden state this as a reason. As to "not usual" there are also wide variations. About 17 per cent of women in Denmark state this as a reason.¹³

Insert Figure 2 here

In contrast, in Sweden and the Netherlands this factor plays no significant role for not getting regular mammograms. "No information about this type of care" plays an important role in Greece (14.6 per cent) and Poland (4.6 per cent) for not undergoing mammograms. Financial restrictions are most important in Greece (2.8 per cent), Germany (2.0 per cent) and Poland (2.4 per cent).

To summarize, there are enormous differences in mammography take-up between European countries (Figure 1) and the individual (Table 1) and institutional factors (Table 2) determining screening takeup as well as the reasons for not undertaking mammogram screening (Figure 2) vary strongly between them.

4 Estimation strategy

The goal of this study is to measure the reasons for differences in mammography screening rates between European countries. Special focus is put on the relative importance of individual versus institutional factors in explaining country differences in screening uptake. Therefore our basic

¹³ The question in SHARELIFE is "What are the reasons you [have never had/stopped having] mammograms regularly?" providing following possible answers: 1. Not affordable, 2. Not covered by health insurance, 3. Did not have health insurance, 4. Time constraints, 5. Not enough information about this type of care, 6. Not usual to get this type of care, 7. No place to receive this type of care close to home, 8. Not considered to be necessary.

regression models are linear multilevel models (MLM).¹⁴ MLM enable us on the one hand to identify the individual and institutional determinants of mammography screening uptake. On the other hand MLM make it possible to analyse whether the differences in individual and institutional factors across countries can explain country differences in screening rates.¹⁵

We define the following two-level random-intercept model:

$$y_{ij} = \beta_0 x_0 + \beta' x_{ij} + \nu_i + \varepsilon_{ij}, \qquad (1)$$

where y_{ij} is whether a woman j in country i gets regularly (i.e. every two years) a mammography screening, x_0 is a constant, x_{ij} is a vector of the explanatory variables and β_0 and β are coefficient estimates. All variables are centered around zero by subtracting their grand mean from the individual values and thus the parameter β_0 can be interpreted as the mean intercept (i.e. screening ratio) across all individuals in all countries. The model has two error components: v_i is the random error for the *i*th country. This error term captures unobserved (i.e. factors that are not controlled for) country specific factors like institutional differences in health system regulation, financing and the provision that cause differences in screening uptake across countries. ε_{ij} is an individual-level random error for the *j*th woman within the *i*th country.¹⁶ This error term captures unobserved individual factors like preferences for prevention, genetic factors, time preferences, exposure to screening when a friend is affected by breast cancer, etc. that might explain individual differences in screening uptake within a country.

We estimate seven variations of equation 1. The first model (Model 1) includes only an intercept mirroring average screening rates across countries. This model enables us to estimate the intra cluster correlation coefficient from the unconditional model as will be explained in greater detail below.

¹⁴ We also estimated a multilevel logit model due to the binary nature of the dependent variable. The marginal effects are very similar to the estimated coefficients of the OLS regressions. Therefore, and because computation is much easier in the linear model, we constrain the presentation of results to the linear probability model. It turns out that depending on the specification, between 3.8 (Model 2; 309 obs.) and 2.7% (Model 7; 437 obs.) of all observations have a predicted value outside the range of [0,1]. We feel that this is a reasonably low figure. ¹⁵ For a nice overview of multilevel models and the application of these models in health economics see Rice and Levland (1996) or Rice and Jones (1997).

¹⁶ By assumption the both error terms have zero mean and constant variances ($\sigma_{\nu}^2, \sigma_{\varepsilon}^2$).

Model 2 adds individual and Model 3 institutional explanatory variables. Models 4 to 7 add the different reasons for not participating in regular mammography screening as explanatory variables. The rationale behind this strategy is to identify in a first step the individual (Model 2) and institutional (Model 3) determinants of mammography uptake within a country ("Analysis of determinants") as well as the "channels" through which individual and institutional variables determine screening uptake (Models 4 to 7).¹⁷

In a second step ("Analysis of variance") this strategy enables us to analyse whether differences across countries in the individual and institutional screening determinants are related to screening differences across countries using the same sequential estimation strategy as above. To this aim we analyse the importance of the country level error v_i term in equation 1 and whether the error term decreases after individual (Model 2), institutional (Model 3) and the other controls (Models 4 to 7) are included in the equation. The idea behind analyzing the change in the magnitude of the country specific error term for explaining country differences is the following: As explained above, the error term captures unobserved country specific factors like institutional differences in health system regulation, financing and providing that cause differences in screening uptake across countries. The bigger the differences in screening rates across countries, the higher the value of the error term will be. The relative decrease in the error term as more individual and institutional controls are added to the model indicates the model's ability to statistically explain cross country differences. Moreover, to identify the relationship between country differences in screening rates and the different reasons for not participating in screening, we sequentially add these different reasons to the model. Finally, to understand the ability of the models to explain country differences even better, we calculate additionally the intra-class correlation coefficients (ICC) as a relative indicator of country variation in screening rates:

¹⁷ Including the reasons for not participating in screening sequentially helps for example to identify the relationship between organised screening and the different reasons for not participating in regular mammography screening. This identification is done by analysing the change of the coefficients of the individual and institutional variables after including the reasons for not participating as additional controls and is explained in more detail in the result section. This could be alternatively analysed by using the different reasons for not participating as outcome variables and regressing them on all other explanatory variables. However, we decide to insert these variables as explanatory variables, since we try to analyse in the following "Analysis of variance" how country differences in the reasons for not undertaking a mammogram are associated with country differences in screening rates after controlling for individual and institutional factors (i.e. we want to figure out what factors drive remaining country differences in screening uptake).

$$\mu_{\nu} = \sigma_{\nu}^2 / (\sigma_{\nu}^2 + \sigma_{\varepsilon}^2)$$
⁽²⁾

The coefficient μ_v defines the proportion of variability in screening rates attributable to the level of the countries and is bounded between 0 and 1. The closer the ICC is to 0, the smaller the proportion of the total variance at country level and the lower the relevance of country specific factors (e.g. institutional factors) for differences in screening rates. In contrast, large values of μ_v indicate that differences in screening rates are attributable to country specific factors.

5 Results

Baseline results

Analysis of Determinants

Table 3 presents the results of the analysis of screening determinants. The results of Model 1, which is used as the baseline model, indicate that the estimated mean screening rate across all countries is 63 per cent. Model 2 adds individual factors as explanatory variables. Almost all individual controls are significant.

Insert Table 3 here

More precisely, a younger age (reference category women aged 50 to 69 years), better health (as measured by sah and adl), better education (i.e. higher ISCED, reference category ISCED 5 to 6) and better cognitive abilities (verbal fluency), having a partner, higher income, not working, a healthy behaviour (i.e. visit a dentist for prevention issues) and taking no medications (i.e. lower risk group) are consistently associated with higher rates of screening take-up.¹⁸

¹⁸ For example, a woman who regularly visits a dentist for preventive issues has a 9.9 percentage points higher probability of getting regularly a mammogram compared to a woman who does not visit a dentist for preventive issues. However, the interpretation of these variables is not of major interest in this paper, since we are interested only in general relevance of individual factors and whether differences in these are relevant for explaining country differences in screening rates. More details results of screening determinants are presented in Wübker (2011).

Model 3 adds institutional variables. The probability that a woman gets a mammogram increases by 38.7 percentage points if a region provides an organized population based screening program. Given the average screening rate of 63 per cent, this is a remarkable amount. Moreover, in countries where a screening program is offered to women outside the age group of 50 to 69 years, the screening probability is on average 15 percentage points higher. After controlling for self-stated financial restrictions, other access problems (e.g. not near available) (Model 4) and the reason "lack of information" (Model 5) the coefficients of the individual and institutional explanatory variables decrease slightly. For example, the coefficient of the organized screening program drops from 0.387 to 0.370 (i.e. by about 4 per cent).¹⁹ This means statistically that the dummy for organized screening is negatively correlated with the variables included in Models 4 and 5 (i.e. financial restrictions, access and information problems). With regard to interpretation, organized screening programs seem to reduce financial restrictions as well as access and information problems. However, including the variables "not usual" (Model 6) and especially "not necessary" (Model 7) as additional controls have a much bigger impact on the coefficients of the institutional explanatory variables. For example the coefficient of an organized screening program drops statistically significant by about 50 per cent from 0.37 (Model 5) to 0.19 (Model 7) indicating that the availability of an organized screening program strongly reduces the probability that a woman believes a screening is "not necessary". Moreover, the coefficients of many individual variables drop sharply towards zero (e.g. education, age, having a partner and income drop between 40 and 60 per cent, the decrease is statistically significant for the age categories, income and "ISCED 1 to 2"- category) and partly lose significance. This result suggests that belonging to the elderly, having low income, being poor educated and not having a partner is strongly related to believing that screening is not necessary.

In summary, individual and institutional factors can statistically explain screening uptake within a country. Quantitatively, the most important factor is the provision of an organized screening program and a central mediating factor for most of the individual and institutional variables is whether a woman thinks that the screening is necessary or not. However, the question remains, to what extent

¹⁹ Note, this decrease is not statistically significant.

differences in individual and institutional factors across countries can statistically explain country differences in screening rates. This question will be answered in the following section.

Analysis of Variance

As described in section 4, a unique feature of multilevel regression analysis is the ability to partition the variance of the dependent variable at different levels (here country versus individual). Table 4 presents two absolute (standard deviation and variance) and one relative (ICC) indicator of variation in screening rates for the different models. For the sake of simplicity we focus on the interpretation of the standard deviation and the ICC. Model 1 with no explanatory variables reveals a standard deviation in screening rates across countries of 19.5 per cent. After controlling for individual factors (Model 2), the standard deviation in screening rates between countries increases slightly by 1.3 percentage points to 20.8 per cent. Thus country differences in individual factors (i.e. age, sah, etc.) cannot explain country differences in screening rates but even lead to increased unexplained differences. Individual factors that are positively associated with screening rates seem to be more prevalent in countries with lower screening rates. By contrast, unexplained country differences in screening rates decrease sharply after controlling for institutional factors (e.g. organized screening program). More precisely, the standard deviation drops by 7.94 percentage points from 20.86 to 12.92 (i.e. in relative values by 38 per cent). The impact of the further inclusion of the selfstated reasons for not participating in screening to the regression model is analysed in Models 4 to 7. Stated financial restrictions and access problems (Model 4), "lack of information" (Model 5) and "not usual" (Model 6) cannot additionally explain differences in screening rates. In contrast, controlling for "not necessary" has an additional enormous impact on the unexplained standard deviation in screening rates between countries. Even after institutional and individual variables are controlled for, the standard deviation decreases by additional 6.19 percentage points from 12.8 to 6.61 (i.e. in relative values by 52 per cent). Turning to the ICC results show that in the model with no controls 21.3 per cent of the overall variance can be partition to the country level, this increases to 23.5 per cent after controlling for individual factors and drops sharply to 9.4 per cent when controlling for institutional factors and drops further to 3.3 per cent when including all reasons for not participating.

Insert Table 4 here

Figure 3 visualises the additional explanatory power of each model and summarizes the three striking results:²⁰ First, observed individual factors cannot explain country differences. Secondly, observed institutional factors explain most of the country differences. For example, the higher screening rate in countries like Spain, the Netherlands, France or Sweden (compared to average screening rate across countries) is almost completely statistically explained by institutional factors (i.e. the availability of an organized screening program and whether a screening program is offered to women outside the age group of 50 to 69 years). Thirdly, "not necessary" is the most important factor for explaining country differences even after individual and institutional factors are controlled for. For instance perceptions and attitudes to screening as measured by "not necessary" (Model 7) seem to be a big barrier in Denmark explaining much of the under-screening in this country (compared to the other countries).

Insert Figure 3 here

Results of sensitivity analysis

²⁰ Note that each variable is centered around zero and thus Figure 3 presents average country screening rates under the assumption that individual variables (Model 2), institutional variables (Model 3), etc. would not be different across countries.

across countries.²¹ Results are presented in the bottom of Table 4. In general, allowing for a more flexible form using random slopes is associated with slightly smaller standard errors and variances. Thus, adjusting for random slopes explains country differences slightly better than not allowing for random slopes. However, in general, results remain robust if we allow for random slopes.²²

Another restrictive assumption within the multilevel model framework is that the country specific effects (error terms) are uncorrelated with the independent variables. This assumption can be relaxed by a country fixed effects regression approach allowing for possible correlation of the country specific effects with the independent variables. Thus we re-estimate Models 1 to 7 by including a dummy variable for each country in equation 1 instead of inserting a country specific random error term. However, a problem of the country fixed effects regression approach is that we cannot control for institutional variables that vary across countries, but display no within country variability (i.e. Extended Age Range). Thus, Table 5 presents the results of fixed effects regression without controlling for country level institutional variables, i.e. for whether a country provides screening to an age range beyond 50 to 69 years. The results can be summarized as follows: First, the coefficients of the individual and institutional explanatory variables are quite similar to those in the multilevel models. Secondly, the differences between the country dummies decrease much after controlling for i) an organized screening program and ii) for "not necessary" as reason for not participating in screening. This is shown by the variances and standard deviations of country dummies presented in the bottom of Table 5. More precisely we calculate the variance (Var) of country dummies as follows:

$$\operatorname{Var} = \frac{1}{13} \sum_{i=1}^{13} (di - d)^2, \qquad (3)$$

where *13* is the number of countries included, *di* is the dummy of country *i* and *d* is the average of the 13 country dummies as calculated by

$$d = \frac{1}{13} \sum_{i=1}^{13} di .$$
 (4)

²¹ This means that it is assumed that the effect of the explanatory variable varies randomly within the population of countries.

²² The results reveal that the error term for random slopes is significant for some of the individual explanatory variables, meaning that there is considerable variation in these coefficients across the countries. The complete results are available upon request.

The Var for Model 1 is thus a measure of the total differences in screenings rates between countries, while the Var for Models 2 to 7 are measures of the remaining unexplained differences, when taking account of an increasing number of individual and institutional characteristics as well as the different reasons for not participating in regular mammography screening. The results indicate the convergence of the variance of country dummies as we move from Model 1 with enormous differences across countries (variance 355 per cent, standard deviation 18.85 per cent) to Model 3 with much lower differences (variance 202 per cent, standard deviation 14.24 per cent) to Model 7 with only minor differences (variance 53 per cent, standard deviation 7.34 per cent) in screening rates across countries. Thus, in general, the results found are also robust to these changes in specification.

Insert Table 5 here

Finally, we tried alternative institutional variables (see Table 2 for these variables) for explaining country differences in screening rates. Since these variables are only available on country level and we have only 13 countries we cannot put them into the models simultaneously. The results are presented in Table A2 of the appendix. In the top of the table, results are presented without controlling for whether an organized screening program exists. In the bottom of the table we control for the availability of an organized screening program. We find a strong statistically significant positive relation between screening uptake and "Age Rate Extended" as well as "Personally Invited". Countries that offer screening uptake rate compared to countries that do not. Moreover, countries that have a 10 percentage point higher invitation rate had a 6.2 percentage point's higher screening uptake rate. The results change strongly if we control for an organized screening program and both variables. The association between the availability of an organized screening uptake drops by 50 per cent but still remains significant and the relation between invitation rate and screening uptake vanishes. The other institutional variables (Mammogram units, Gatekeeper, Fee for Service etc.) show, with and without

controlling for an organized screening program, no statistical significant relation to screening uptake and hardly influence the coefficient of organized screening demonstrating the dominant role of organized screening in explaining mammography decision.²³ Lastly, we analyzed the relation between screening uptake and breast cancer incidence as well as breast cancer mortality. Without controlling for organized screening we find a positive correlation between screening and incidence and a negative correlation between screening and mortality. However, the correlation between incidence of breast cancer and screening uptake turns out to be negative if we control for organized screening indicating a positive association between breast cancer incidence and the availability of an organized screening program.

6 Discussion and Conclusion

Breast cancer is the main cause of cancer-mortality among women in Europe. Screening mammography helps to detect breast cancer before it becomes invasive, and mortality can be significantly reduced by regular mammography screening. Even though mammography is officially recommended both on the national and European level for women aged 50 to 69, screening rates in most European countries remain far from 100 per cent and differ much between European countries. Due to lack of comparable cross national data no study exists that analyses the statistical causes of screening differences across countries and the fundamental reasons for these variations remain largely unexplored and not well understood. Against this background the purpose of this paper was to explore the statistical causes of these country differences in screening rates of woman aged 50-69 using data from the SHARE. Since differences in screening rates across countries might stem from differences in individual or institutional factors (Bolin et al. 2009), special focus is put on the relative importance of individual versus institutional factors in explaining country differences in screening uptake.

The results indicate that observed individual factors (i.e. age, education, etc.) can statistically explain within country variation in screening rates but cannot explain between country variations in screening

²³ This makes sense, since the supply of an organized screening program mirrors a comprehensive public health intervention influencing incentives on both the demand and supply side as described in section 2.

rates. Further, observed institutional factors (i.e. the availability of an organized screening program and whether screening is provided beyond the age group of 50 to 69 years) can statistically explain about 40 per cent of the between country differences in screening rates. The impact of these institutional factors seem to be mediated to a minor extent (by about 4 per cent) by solving financial restrictions, access problems as well as "lack of information" and to a major extent (by about 50 per cent) by avoiding that a woman believes a screening is "not necessary". This result might indicate that the availability of a screening program increases perceived benefits of screening reflecting that national screening programs have reduced mortality over time (Kalager *et al.*, 2010). It also might resolve concerns about false positive treatments, the treatment of potentially benign tumours or identifications of mainly low risk cancers,²⁴ since there is some evidence that organized screening protects better against harms compared to traditional opportunistic screening.²⁵ Thirdly, and additionally to the indirect association of "not necessary" with individual and institutional factors, differences across countries in whether a woman believes that screening is not necessary can statistically explain an additional 50 per cent of the between country differences.

Policy Implication

In terms of health policy implications the results found suggest that the availability of an organized screening program (i.e. differences in the institutional framework) strongly matters for explaining current screening differences across European countries. Since it is not clear to what extent people have an understanding of harms (e.g. overtreatment, anxiety associated with false positive results) associated with screening when weighing up the costs and benefits of screening, this analysis can provide some insights about patient attitudes and understanding and it highlights the importance of the availability of an organized screening program for individual screening decisions. Moreover, the analysis reveals that, if policy intends to increase screening rates, changes in the institutional setting

²⁴ There might be a reduction in quality of life due to false positive findings and possibly unnecessary treatment (although the magnitude of the extent or significance of over-treatment is debatable with a number of studies estimating small degrees of over-treatment). Compare Raftery and Chorozoglu (2011) and Hackshaw (2012) on a discussion of these issues. But even the quality of life decrease from treatment may not be that large if patients are convinced they have successfully treated a life threatening cancer.

²⁵ Compared with opportunistic screening, organized screening focuses much greater attention on the quality of the screening process, including follow-up of participants. Compare Miles et al. (2004).

might in particular address the perceived necessity of screening. Since patients both (a) often follow physician recommendation and (b) provider counselling about screening is more important for screening decision than poor patient acceptance (e.g. Haas *et al.*, 2007), public health interventions might address physicians recommendations on screening. Future research might investigate which public health interventions (e.g. pay for performance) might be most cost-effective in influencing the decision to recommend a screening by a physician.

Literature

- Autier P, Ouakrim DA. 2008. Determinants of the number of mammography units in 31 countries with significant mammography screening. *British Journal of Cancer* **99**: 1185 1190.
- Anthony DL, Brooke Herndon M, Gallagher PM, Barnato AE, Bynum JPW, Gottlieb DJ, et al. 2009. How much do patients' preferences contribute to resource use? *Health Affairs* **28**(3): 864-873.
- Biesheuvel C, Weigel S, Heindel W. 2011. Mammography screening: evidence history and current practice in Germany and other European countries. *Breast Care* 6:104–1096.
- Börsch-Supan A, Jürges H. 2005. The survey of health, aging and retirement in Europe methodology. *Technical Report*, Mannheim Research Institute for the Economics of Aging, Mannheim.
- Bolin K, Lindgren A, Lindgren B, Lundborg P. 2009. Utilisation of Physician Services in the 50+ Population. The Relative Importance of Individual versus Institutional Factors in 10 European Countries. *National Bureau of Economic Research Working Paper No.* **14096**.
- Byrne MM, Thompson P. 2001. Screening and preventable illness. Journal of Health Economics **20** (6): 1077–1088.
- Carrieri V, Bilger M. (2013). Preventive care: underused even when free. Is there something else at work? *Applied Economics*, **45**(2): 239-253.
- Chandra A, Cutler D, Song Z. 2012. Who ordered that? The economics of treatment choices in medical care. In: Pauly MV, McGuire TG, Barros PP. (Eds.), *Handbook of Health Economics*, Vol.2, Chapter 3: 46-93, Amsterdam: Elsevier (North-Holland).
- Chang FR. 1996. Uncertainty and investment in health. Journal of Health Economics 15: 369-376.
- Commission of the European Communities COM. 2008. Report from the commission to the council. The European Parliament, The European Economic and Social Committee and the Committee of the Regions. (Available from:
 - http://ec.europa.eu/health/ph determinants/genetics/documents/com 2008 882.en.pdf)

- Cropper M. 1977. Health, investment in health, and occupational choice. *Journal of Political Economy* **85:** 1273–1294.
- Dardanoni V, Wagstaff A. 1990. Uncertainty and the demand for medical care. *Journal of Health Economics* **9**: 23 -38.
- European Observatory on Health Systems and Policies. 2012. Health Systems in Transition (HiT) series. (Diverse country reports avaiable from: http://www.euro.who.int/en/who-we-are/partners/observatory/health-system-reviews-hits/full-list-of-hits)
- Fang H, Wang J. 2010. Estimating dynamic discrete choice models with hyperbolic discounting, with an application to mammography decisions. *National Bureau of Economic Research Working Paper* No **16438**.
- Ferlay J, Autier P, Boniol M, Heanue M, Colombet M, Boyle P. 2007. Estimates of the cancer incidence and mortality in Europe in 2006. *Annals of Oncology* **18**: 581-592.
- Giuffrida A, Gravelle H. 1998. Paying patients to comply: an economic analysis. Health Economics 7 (7): 569– 579.
- Giorgi D, Zappa M, Giordano L, Ventura L, Frigerio A, Paci E. 2007. Mammography screening in Italy: 2004 survey and 2005 preliminary data. *Epidemiologia E Prevenzione* **31**(2-3) Suppl 2: 7-20.
- Gøtzsche PC, Nielsen M. 2009. Screening for breast cancer with mammography. *Cochrane Database* of Systematic Reviews 7 (4):CD001877.
- Grossman M. 1972. On the concept of health capital and the demand for health. *Journal of Political Economy* **80**(2): 223–55.
- Hackshaw A. 2012. Benefits and harms of mammography screening; Using evidence from randomised and observational studies is necessary and appropriate. *British Medical Journal* **344**:d8279 doi: 10.1136/bmj.d8279.
- Haas JS, Fitzmaurice G, Brawarsky P, Liang SY, Hiatt RA, Klabunde CN, Brown ML, Phillips KA.
 2007. Association of regional variation in primary care physicians' colorectal cancer screening recommendations with individual use of colorectal cancer screening. *Preventing Chronic Disease* 4 (4): A90. (Available from:

http://ukpmc.ac.uk/articles/PMC2099288//reload=0;jsessionid=dOP2kyZUr2MrzfFS2lvz.16)

- Hamblin JE. 1991. Physician recommendations for screening mammography: results of a survey using clinical vignettes. *The Journal of Family Practice* **32**(5), 472–477.
- International Agency for Research on Cancer IARC. 2005. Cervix cancer screening. *IARC Handbook of cancer prevention*. Vol 10. IARC Press, Lyon.
- International Agency for Research on Cancer IARC. 2002. Breast Cancer Screening. *IARC Handbooks of Cancer Prevention*. Vol. 7. IARC Press, Lyon.
- Jepson R, Clegg A, Forbes C, Lewis R, Snowden A, Kleijnen J. 2000. The determinants of screening uptake and interventions for increasing uptake: a systematic review. *Health Technology Assessment* **4** (14), i–vii: 1–133.

- Jørgensen KJ, Zahl PH, Gøtzsche PC. 2010. Breast cancer mortality in organised mammography screening in Denmark: comparative study. *British Medical Journal* **340**:c1241. (Available from: http://www.bmj.com/content/340/bmj.c1241.full)
- Kalager M, Zelen M, Langmark F, Adami HO. 2010. Effect of Screening Mammography on Breast-Cancer Mortality in Norway. *New England Journal of Medicine* 363: 1203-10.
- von Karsa L, Anttila A, Ronco G, Ponti A, Malila N, Arbyn M, Segnan N, Castillo-Beltran M, Boniol M, Ferlay J, Hery C, Sauvaget C, Voti L, Autier P. 2008. Cancer screening in the European Union. *Report on the implementation of the Council Recommendation on cancer screening First Report*. European Communities (publ.). (Available from: URL: http://ec.europa. eu/health/ph determinants/genetics/documents/ cancer screening.pdf.)
- Kooperationsgemeinschaft Mammographie: Evaluationsbericht 2008–2009. 2012. Ergebnisse des Mammographie-Screening-Programms in Deutschland. Berlin.
- Lairson DR, Chan W, Newmark GR. 2005. Determinants of the demand for breast cancer screening among women veterans in the United States. *Social Science & Medicine* **61**(7): 1608-1617.
- Legler JM, Meissner HI, Coyne C, Breen N, Chollette V, Rimer BK. 2002. The effectiveness of interventions to promote mammography among women with historically lower rates of screening. *Cancer Epidemiology, Biomarkers & Prevention* **11** (1): 59-71.
- Levin TR, Jamieson L, Burley DA, Reyes J, Oehrli M, Caldwell C. 2011. Organized colorectal cancer screening in integrated health care systems. *Epidemiologic Reviews* **33** (1): 101-110.
- Li J, Hurley J, DeCicca P, Buckley G. 2011. Physician Response to Pay-for-Performance: Evidence from a Natural Experiment. *National Bureau of Economic Research Working Paper* No. 16909.
- Mandelblatt JS, Yabroff KR. 1999. Effectiveness of interventions designed to increase mammography use: a meta-analysis of provider-targeted strategies. *Cancer Epidemiology, Biomarkers & Prevention* **8**: 759–67.
- McGuire T. 2000. Physician agency. In: Culyer A, Newhouse J. (Eds.), *Handbook of Health Economics*, Vol. 1: 461-536, Amsterdam, Elsevier.
- Miles A, Cockburn J, Smith RA et al. 2004. A perspective from countries using organized screening programs. *Cancer* 101(suppl 5):1201–1213.
- Moore SG, Shenoy PJ, Fanucchi L, Tumeh JW, Flowers CR. 2009. Cost-effectiveness of MRI compared to mammography for breast cancer screening in a high risk population. *BMC Health Services Research* **13**: 9:9. (Available from: http://www.biomedcentral.com/1472-6963/9/9)
- National Cancer Institute; International Cancer Screening Network. 2012. Organization of breast cancer screening programs in 27 ICSN countries, 2007-2008. (Available from: http://appliedresearch.cancer.gov/icsn.)
- O'Hare AM, Rodriguez RA, Hailpern SM, Larson EB, Kurella Tamura M. 2010. Regional variation in health care intensity and treatment practices for end-stage renal disease in older adults. *Journal of the American Medical Association* **304**(2): 180-186.

- Organisation for Economic Co-operation and Development OECD. 2011. OECD Health Data 2011. OECD, Paris. (Available from: http://stats.oecd.org/index.aspx?DataSetCode=HEALTH_STAT)
- Organisation for Economic Co-operation and Development OECD. 2009. Health at a Glance 2009 OECD Indicators. OECD, Paris. (Available from: http://www.oecdilibrary. org/content/book/health_glance-2009-en.)
- Peek ME, Han JH. 2004. Disparities in screening mammography. Current status, interventions and implications. *Journal of General Internal Medicine* **19**(2): 184-194.
- Quanstrum KH, Hayward RA. 2010. Lessons from the mammography wars. New England Journal of Medicine 363:1076-1079.
- Raftery J, Chorozoglou M. 2011. Possible net harms of breast cancer screening: updated modelling of Forrest report. *British Medical Journal* **343**:d7627. (Available from: http://www.bmj.com/content/343/bmj.d7627.short?rss=1)
- Rice N, Jones A. 1997. Multilevel models and health economics. Health Economics 6: 561-575.
- Rice N, Leyland R. 1996. Multilevel Models: Applications to health data. *Journal of Health Services Research & Policy* 1: 154–164.
- Schmitz H, Wübker A. 2011. What determines flu vaccination take up of elderly Europeans? *Health Economic* **20**: 1281-1297.
- Schopper D, de Wolf C. 2007. Breast cancer screening by mammography: International evidence and the situation in Switzerland, Krebsliga Schweiz. (Available from: http://assets.krebsliga.ch/downloads/short_mammo_report_final_el_.pdf)
- Schueler KM, Chu PW, Smith-Bindman R. 2008. Factors associated with mammography utilization: a systematic quantitative review of the literature. *Journal of Womens Health (Larchmt.)* **17** (9): 1477-1498.
- Skinner J. 2012. Causes and consequences of regional variations in health care. In: Pauly MV, McGuire TG, Barros PP. (Eds.), *Handbook of health economics*, Vol.2, Chapter 3: 46-93, Amsterdam: Elsevier (North-Holland).
- Strumpf EC, Zhijin C, Kadiyala S. 2010. Adherence to cancer screening guidelines across Canadian provinces: an observational study. *BMC Cancer* 10: 304. (Available from: http://www.biomedcentral.com/1471-2407/10/304)
- Wübker A. 2011. Who gets a mammogram amongst European women aged 50-69 years and why are there such large differences across European countries? *Beiträge zur Jahrestagung des Vereins für Socialpolitik* 2011: Die Ordnung der Weltwirtschaft: Lektionen aus der Krise - Session: Health Economics, No. F16-V1.
- Yasmeen S, Romano PS, Tancredi DJ, Saito NH, Rainwater J, Kravitz RL. 2012. Screening mammography beliefs and recommendations: a web-based survey of primary care physicians. *BMC Health Services Research* 12:32. (Avaiable from: http://www.biomedcentral.com/1472-6963/12/32)

Insert Table A1 here

Insert Table A2 here

Table 1: Descriptive Statistics for women aged 50 to 69 years (average of 2004 and 2006)

	M		CAT.		Hist.	ISCED	Verbal	Having	Log	Work-	Dent	C	No
	Mamm.	Age	SAH	ADL	BC	ISCED	Fluency	Partner	Income	ing	Prevent.	Smoker	Medic.
	0.00			0.00			22.04	0.54		0.15		0.10	
Austria	0.68	64.44	3.16	0.09	0.023	2.75	22.04	0.56	2.66	0.17	0.44	0.19	0.27
Germany	0.48	63.19	3.25	0.08	0.026	3.41	22.64	0.80	2.92	0.38	0.73	0.14	0.35
Sweden	0.82	63.97	2.83	0.10	0.024	3.14	25.17	0.76	3.02	0.54	0.74	0.21	0.39
Netherlands	0.86	62.48	3.01	0.09	0.022	2.84	21.61	0.78	2.92	0.34	0.70	0.26	0.40
Spain	0.78	63.01	3.60	0.12	0.011	1.59	15.44	0.81	2.26	0.22	0.21	0.12	0.26
Italy	0.66	63.38	3.40	0.08	0.020	1.90	15.65	0.81	2.42	0.18	0.24	0.16	0.28
France	0.85	62.57	3.19	0.09	0.023	2.61	21.59	0.61	2.84	0.41	0.29	0.14	0.25
Denmark	0.26	61.80	2.75	0.08	0.038	3.65	24.00	0.68	3.32	0.54	0.84	0.27	0.43
Greece	0.43	62.43	2.91	0.08	0.016	2.23	15.08	0.63	2.28	0.24	0.34	0.27	0.25
Switzerland	0.49	62.59	2.66	0.07	0.019	2.84	22.62	0.66	3.39	0.50	0.67	0.22	0.46
Belgium	0.80	62.60	3.03	0.12	0.037	2.92	21.20	0.72	0.03	0.30	0.47	0.16	0.25
Cz. Republ.	0.65	62.01	3.35	0.07	0.013	2.49	20.53	0.68	1.73	0.29	0.53	0.21	0.26
Poland	0.43	61.05	3.83	0.35	0.012	2.39	16.72	0.75	1.21	0.20	0.13	0.26	0.29
Total	0.64	62.73	3.14	0.11	0.023	2.69	20.27	0.72	2.65	0.33	0.49	0.20	0.31
SAH = Self A	ssessed He	ealth; AI	DL = N	umber c	of limita	tions in A	ctivities of	Daily Livi	ng; ISCEI	D = Intern	ational Sta	ndard	
Classification	of Educati	on (leve	e1 0 = Pr	e-prima	ry educa	ation to lev	vel 6 = Sec	ond stage	of tertiary	educatior	ı), Hist. BC	= History	of
Breast Cancer													

factor
(institutional)
d other
ns and
prograi
screening
0 OI
statistics
Descriptive
Table 2:

		Measures of intens	f screening sity									Incid.	Mortality
	Organized	Personally invited of		Year first	Current		Number of Radiologists	Number of				Breast Cancer	Breast Cancer
	screening	target	Screening	screening	(2006)	Age Range	per million	Mammography	GP as		Doctors	per	per
	availability 1,2	population in 2006 ^{1,2}	interval (vears) ²	program started ²	Program Tvne ²	Extended (vears) ²	women after 2002 ³	units per million women (vear) ³	Gate- keeper ⁴	Type of Health Care Svstem ⁴	type of pavment ⁴	100,000 in 2004^{5}	100,000 in 2004 ⁵
Austria	0.00	0.00	(1) 2	1999	Org. S., Reg; Opp. S., N	1 (40-69)	226.19	100 (2005)	No	Social insurance decentralized		91.5	24.6
Junio	0.00	50.0	, , , , , , , , , , , , , , , , , , , ,	2006	Org. S., N;	0720 60)	20 07 1	51 (2001)	No	Social insurance	ŭ	¢ 1¢1	5 96
Germany	0.00	C0.0	4	C007	Oro S N.	(60-0C) N	1+7.41	(1007) 10	ONI	Tax financed	4	7.171	C.07
Sweden	1.00	0.67	1.5	1986	Opp. S., N	1 (40-74)	214.07	38 (2006)	Yes	decentralized	S	125.8	21.1
Netherlands	1.00	0.47	2	1989	Org. S., N; Opp. S., N	1 (50-75)	101.10	20 (2005)	Yes	Social insurance decentralized	С	122.6	29.8
Spain	1.00	0.44	2	1990	Org. S., N; Opp. S., N	1 (45-70)	177.77	52 (2004)	Yes	Tax financed decentralized	s	93.6	19.2
Italy	0.65	0.27	2	2002	Org. S., Reg; Opp. S., N	0 (50-69)	334.56	86 (2005)	Yes	Tax financed decentralized	C	105.3	24
France	1.00	0.52	2	2002	Org. S., N; Opp. S., N	1 (50-74)	238.22	87 (2006)	No	Social insurance	Ч	127.4	25.9
Denmark	0.22	0.08	2	1991	Org. S, Reg.; Opp. S., N	0 (50-69)	383.21	20 (2003)	Yes	Tax financed decentralized	С	122.6	34.5
Greece	0.00	0.00	(1) 2	1989	Opp. S. N	0 (40/50-64)	442.48	72 (2005)	No	Tax financed	S	81.8	21.7
Switzerland	0.41	0.125	2	1999	Org. S, Reg.; Opp. S., N	0 (50-69)	174.87	79 (2005)	No	Priv. financed decentralized	F	126.5	22.8
Belgium	1.00	0.50	2	1994	Org. S, Reg.; Opp. S., N	0 (50-69)	276.08	55 (2006)	No	Social insurance	F	137.8	33.5
Cz. Republic	0.00	0.00	2	2002	Opp. S, N	1 (45-69)	246.76	28 (2003)	No	Social insurance decentralized	C	93.1	26.7
Poland	0.00	0.00	2	2007	Org. S, N	0 (50-69)	121.21	23 (2005)	Yes	Social insurance decentralized	С	74.1	20.9
¹ A value of regional vari these regions of Danish w(Karsa <i>et al.</i> (Kooperations Screening; O Sources: ¹ Sc Ouakrim (20)	zero indicates ti ation in screenir a dummy varia omen included i 2008); Detailed specenischaft M ppger S. = Opport hopper and de N 08); ⁴ Country re	at no organized s ag programs within bble and calculated in the SHARE dati regional informat Aammographie (2 unistic Screening, Wolf (2007); ² von Ports of European	screening program of the share of d the share of d the share of d the share of d the other of d the other of d of d the other of d o	or example in vomen that h vomen that h vomen that h vomen an organized vise with regic vide, Reg = Re vide, Reg = Re vide, and Nav	ole and a value of Demnark only in Denmark only and access to an d screening proj onal variation re an 2007 (Italy), egional tional Cancer In tional Cancer In stems and Polici	of 1 signals that i Kopenhagen-cit organized screer gram. General in 'garding the impl Jorgensen, Zahl, Jorgensen, Zahl, stittute; Internati, zies, see http://w/	an organized scru y, Funen and Fru ining program bas formation on the ementation of sc , Gørtzsche (2010 , Gørtzsche (2010 , onal Cancer Scru zww.euro.who.int	cening program is cering provid eed on SHARE-dat is implementation st (Denmark); Schl Denmark); Schl zening Network (2) ; Biesheuvel, Weig ; Biesheuvel, Weig	nationwidd ed an orga a. Thus thu atus of org were prov nopper, de 012), see l	e available. Value mized screening p e value of 0.22 fon ganized screening ded by following Wolf (2007) (Swi http://appliedresea	s between 0 cogram until cogram until programs w sources: itzerland); C rch.cancer. <u>f</u> fay <i>et al.</i> (2	and 1 point 1 2006. We neans that 2 'ere derived Jrg. S. = Or Jrg. S. = Or (006).	to set for 2 percent from ganized vutier and

	Model 1	Model 2	Model 2	Model 4	Model 5	Model 6	Model 7
$50 \leq -\Lambda q_0 \leq 55$	Widdel 1	0.088***	0.001***	0.002***	0.087***	0.081***	0.048***
50 - Age - 55		(6.36)	(6.74)	(6.88)	(6.52)	(6.17)	(4.10)
$55 \le A \sigma e \le 59$		0.094***	0.094***	0.093***	0.089***	0.085***	0.047***
55 · 11 <u>5</u> 6 · 57		(7.51)	(7.75)	(7,72)	(7.40)	(7.13)	(4 44)
$60 \le Age \le 64$		0.059***	0.059***	0.060***	0.055***	0.052***	0.026*
00 - 1150 - 01		(4.86)	(4.97)	(5.12)	(4 72)	(4 54)	(2.52)
Self-assessed Health		-0.013**	-0.011*	-0.011*	-0.011*	-0.010*	-0.011**
Sell ussessed Health		(-2.84)	(-2, 50)	(-2,51)	(-2.44)	(-2,26)	(-2.81)
ADL		-0.014	-0.016	-0.014	-0.014	-0.017*	-0.014
11012		(-1.65)	(-1.90)	(-1.71)	(-1.70)	(-2.11)	(-1.94)
Hist. Breast Cancer		0.223***	0.203***	0.196***	0.191***	0.186***	0.113***
		(8.39)	(7.81)	(7.62)	(7.44)	(7.34)	(5.03)
ISCED 0		-0.120***	-0.112***	-0.105***	-0.099***	-0.092***	-0.053*
		(-4.17)	(-4.01)	(-3.78)	(-3.57)	(-3.36)	(-2.20)
ISCED 1 to 2		-0.050***	-0.054***	-0.048***	-0.044***	-0.041***	-0.021*
		(-3.92)	(-4.32)	(-3.89)	(-3.58)	(-3.41)	(-2.02)
ISCED 3 to 4		-0.011	-0.013	-0.013	-0.012	-0.011	-0.000
		(-0.91)	(-1.12)	(-1.08)	(-1.06)	(-0.99)	(-0.05)
Verbal_Fluency		0.003***	0.003***	0.002***	0.002***	0.002**	0.001*
		(4.90)	(3.84)	(3.65)	(3.47)	(3.16)	(2.17)
Having_Partner		0.032***	0.035***	0.032***	0.030***	0.029**	0.014
		(3.40)	(3.79)	(3.56)	(3.30)	(3.24)	(1.72)
Log_Income		0.019***	0.014**	0.015***	0.015***	0.014**	0.005
		(4.22)	(3.02)	(3.39)	(3.39)	(3.24)	(1.27)
Working		-0.005	-0.006	-0.002	-0.003	-0.002	0.001
		(-0.45)	(-0.55)	(-0.15)	(-0.28)	(-0.15)	(0.07)
Dentist		0.099***	0.096***	0.093***	0.093***	0.091***	0.061***
		(10.83)	(10.71)	(10.56)	(10.55)	(10.43)	(8.00)
Smoker		-0.024*	-0.029**	-0.030**	-0.028**	-0.026**	-0.014
NY N/ 41 1		(-2.31)	(-2.86)	(-3.00)	(-2.81)	(-2.60)	(-1.64)
No_Medication		-0.070***	-0.072***	-0.071***	-0.073***	-0.073***	-0.048***
0.0		(-7.39)	(-7.67)	(-7.75)	(-7.97)	(-7.98)	(-6.03)
OrgScreen			0.38/***	0.377***	0.370***	0.355***	0.191***
Ent AssBauss			(22.90)	(22.46)	(22.20)	(21.48)	(13.26)
Ext_AgeRange			0.151*	0.146	0.141	0.150	0.075*
Ein Destriction			(2.08)	(1.95)	(1.85)	(1.89)	(1.98)
Fin_Restriction				-0.353***	-0.339***	-0.344****	-0.331***
No. Timo				(-0.37)	(-0.29)	(-0.47)	(-9.60)
N0_1 lille				(11.52)	(10.86)	(10.84)	(12.65)
Not Near				(-11.32) 0.242***	0 1/8**	0.123*	0.046
not_near				(-4.95)	(-3.02)	(-2.53)	(-1.06)
No. Information				(1.55)	-0.306***	-0.236***	-0.180***
Ito_Information					(-11.84)	(-9.03)	(-7.79)
Not Usual					(11.01)	-0 290***	-0 339***
<u>o</u> ouu						(-14 38)	(-19.05)
Not Necessary						(11.50)	-0 536***
							(-57.19)
Constant	0.630***	0.632***	0.646***	0.647***	0.647***	0.646***	0.642***
	(11.57)	(10.88)	(17.73)	(17.16)	(16.96)	(17.92)	(33.96)
Observations	11409	11409	11409	11409	11409	11409	11409
		**	****	11.02	11.07	11.02	

 Table 3: Determinants of mammography screening

t statistics in parentheses, p < 0.05, p < 0.01, p < 0.01

	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6	Model 7		
Variance components (rand	om intercep	ot Model)							
Between countries variance $\delta^2 v(k)$	383***	435***	167***	179***	183***	164***	43.7***		
Between countries standard deviation in per cent δυ(k)	19.57***	20.86***	12.92***	13.37***	13.54***	12.80***	6.61**		
Between individuals variance δ²ε(jk)	1799***	1847***	1767***	1730***	1709***	1679***	1311***		
Between individuals standard deviation in per cent δυ(k)	42.42***	42.98***	42.04***	41.60***	41.34***	40.98***	36.21***		
Intra-cluster correlation (ICC)									
Intra-cluster correlation in per cent (ICC $\delta^2 v(k)$)	21.28	23.56	9.4	10.3	10.7	9.7	3.3		
Variance components (random	slope Model)							
Between countries variance $\delta^2 v(k)$	383***	419***	148***	161***	165***	149***	40.9***		
Between countries standard deviation in per cent δυ(k)	19.57***	20.47***	12.15***	12.67***	12.86***	12.20***	6.40***		
Between individuals variance δ ² ε(jk)	1799***	1815***	1728***	1696***	1679***	1652***	1296***		
Between individuals standard deviation in per cent $\delta v(k)$	42.42***	42.61***	41.58***	41.18***	40.97***	40.64***	36.02		
Intra-cluster correlation (ICC))								
Intra-cluster correlation in per cent (ICC $\delta^2 v(k)$)	21.28	23.08	8.5	9.5	9.8	9.0	3.2		

Table 4: Variance and standard deviation of mammography screening

* p < 0.05, ** p < 0.01, *** p < 0.001

Table 5: Results of countr	y fixed effects regression
----------------------------	----------------------------

	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6	Model 7
Austria	0.680***	0.683***	0.902***	0.907***	0.901***	0.897***	0.734***
Germany	0.470***	0.419***	0.608***	0.607***	0.606***	0.599***	0.626***
Sweden	0.821***	0.768***	0.600***	0.598***	0.596***	0.594***	0.594***
Netherlands	0.861***	0.833***	0.661***	0.657***	0.653***	0.651***	0.674***
Spain	0.789***	0.866***	0.681***	0.679***	0.677***	0.673***	0.673***
Italy	0.661***	0.717***	0.671***	0.672***	0.668***	0.662***	0.672***
France	0.849***	0.863***	0.688***	0.685***	0.681***	0.683***	0.680***
Denmark	0.257***	0.178***	0.320***	0.312***	0.312***	0.348***	0.467***
Greece	0.430***	0.472***	0.684***	0.695***	0.725***	0.733***	0.636***
Switzerland	0.485***	0.448***	0.513***	0.511***	0.508***	0.499***	0.564***
Belgium	0.796***	0.776***	0.602***	0.602***	0.598***	0.598***	0.631***
Czechia	0.663***	0.678***	0.891***	0.883***	0.881***	0.873***	0.767***
Poland	0.428***	0.510***	0.715***	0.731***	0.728***	0.718***	0.693***
50 <= Age < 55		0.088**	0.090***	0.092***	0.086***	0.081***	0.047***
55 <= Age < 59		0.093***	0.094***	0.093***	0.088***	0.084***	0.046***
$60 \le Age \le 64$		0.059***	0.058***	0.060***	0.055***	0.052***	0.025*
Self-ass. Health		-0.013*	-0.011*	-0.011*	-0.011*	-0.010*	-0.011**
ADL		-0.014	-0.016	-0.014	-0.014	-0.017	-0.014
Hist Breast Cancer		0.223***	0.203***	0.197***	0.191***	0.186***	0.113***
ISCED 0		-0.121*	-0.111***	-0.104**	-0.098**	-0.091**	-0.052
ISCED 1 to 2		-0.050	-0.054***	-0.048***	-0.044***	-0.041**	-0.022
ISCED 3 to 4		-0.011	-0.014	-0.013	-0.013	-0.012	-0.001
Verbal_Fluency		0.003*	0.003***	0.002***	0.002***	0.002**	0.001*
Having_Partner		0.032*	0.035**	0.032**	0.030**	0.029**	0.013
Log_Income		0.019*	0.014**	0.016**	0.015**	0.015**	0.006
Working		-0.005	-0.005	-0.001	-0.002	-0.001	0.001
Dentist		0.099***	0.096***	0.094***	0.093***	0.092***	0.063***
Smoker		-0.024	-0.029*	-0.030**	-0.028*	-0.026*	-0.014
No_Medication		-0.070***	-0.071***	-0.071***	-0.073***	-0.072***	-0.048***
OrgScreen			0.392***	0.382***	0.375***	0.361***	0.196***
Fin Restriction				-0.353***	-0.340***	-0.344***	-0.351***
No Time				-0.440***	-0.413***	-0.409***	-0.421***
Not Near				-0.242***	-0.149**	-0.124*	-0.047
No Information					-0.307***	-0.236***	-0.180***
Not Usual						-0.289***	-0.337***
Not Necessary							-0.535***
Observations	11409	11409	11409	11409	11409	11409	11409
Variance Country							
Dummys in per							
cent	355.19	404.24	202.77	208.71	208.85	189.22	53.93
Standard Deviation							
Country Dummys							
in per cent	18.85	20.11	14.24	14.45	14.45	13.76	7.34
p < 0.10, p < 0.05	5, p < 0.01	l					

Variables		Variable description	Mean
Endogenous		•	
		Mamm. screening at least every two	
	Mammogram	years(Yes=1, No=0)	0.64
Exogenous			
Age	50 <= Age < 55		0.25
	55 <= Age < 60		0.28
	$60 \le Age \le 65$		0.25
Health Status	SAH	Self-Assessed Health (excellent =1 to poor = 5)	2.92
	ADL	Number of limitations in Activities of Daily	
	II' (D)	Living	0.11
	Hist, Breast	A history of broast Cancor	0.022
Family Structure	Having Partner	Women has a partner	0.023
Tanniy Structure	ria ing raraier	Logarithmized household income adjusted by	0.72
Income	Log Income	the square root of the household size	2.64
	0_	International Standard Classification of	
Education	ISCED 0	Education (ISCED) level 0	0.027
	ISCED 1 to 2	ISCED level 1 to 2	
	100000 4		0.431
	ISCED 3 to 4	ISCED level 3 to 4	0.242
	Verbal Fluency	Counts the number of different animals the	0.342
	verbai Fluency	respondent is able to	
Cognition		state within 1 min	20.27
Working	Working	Women is working or self employed	0.33
e	Smoking	Women is currently smoking	0.49
	Dent_Prevent	Women visited dentist for routine control or	
Health Behaviour		prevention	0.20
Risk_Group	No_Medication	Women takes not regularly medication	.31
Institutional factors	Organized	One mine d Comming and lable	0.55
(only those included in the main analysis	A ge Range	Screening available to age groups beyond 50 to	0.55
Compare Table 2 for the other ones)	Extended	69	0.40
	Financial Constr	"Too Expensive" or "Not Covered by	0.10
	aint	Insurance" or "No Insurance"	0.009
	Not Available	"No place to receive this type of care close to	
		home"	0.007
	No Time	"No time"	0.011
	No Information	"Not enough information about this type of	0.00
Calf stated managers for not having	Not Liquel	care" "Not usual to get this type of aero"	0.026
mammogram	Not Nacassarr	"Not considered to be pagesony"	0.043
mammogram	Not Necessary	"Not considered to be necessary"	0.196

Table A1: Sample means and description

	Model	Model	Model	Model	Model	Model	Model	Model	Model	Model
	1	2	3	4	5	6	. 7	8	9	10
Org. Screen.	0.390***		Withou	it Controlli	ng for Organ	lized Screen	ing			
Ext_Age Range	(23.00)	0.279***								
Real invited		(3.17)	0.582***							
Mamm. Units			(3.42)	0.002						
Radiologists				(0.70)	-0.001					
Gatekeeper					(-1.18)	0.025				
Fee for Serv.						(0.21)	-0.007 (-0.05)			
Start program								-0.004		
Incidence BC								(-0.49)	0.001	
Mortality									(0.30)	-0.010
BC										(-0.77)
Individual Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
			With	Controlling	for Organi	zed Screenir	ıg			
Org. Screen.	0.390*** (23.06)	0.387*** (22.90)	0.394*** (22.80)	0.390*** (23.05)	0.390*** (23.00)	0.391*** (23.08)	0.391*** (23.06)	0.392*** (23.08)	0.394*** (23.26)	0.390*** (23.09)
Ext_Age Range		(2.08)								
Real Invited		(2.08)	-0.163 (-0.94)							
Mamm. Units				0.001						
Radiologists				(0.98)	-0.000					
Gatekeeper					(-0.75)	-0.090 (-1.10)				
Fee for Serv.						()	-0.077 (-0.86)			
Start program								0.007		
Incidence BC								(1.17)	-0.004*	
Mortality BC									(-2.29)	-0.012
DC										(-1.42)
Individual Controls ¹	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Constant	0.656***	0.646***	0.652***	0.657***	0.653***	0.657***	0.652***	0.655***	0.644***	0.647***

Table A2: The association between screening uptake and alternative institutional controls

 $\frac{11409}{11409} \frac{11409}{11409} \frac{11409}{1140} \frac{11409$

Figure 1: Mammography screening rates of women aged 50 to 69 years in European countries in 2006 based on SHARE data



Figure 2: Self-stated reasons for not participating in mammography screening of women aged 50

to 69 years





Figure 3: Explanatory power of the models